Author’s response to reviews

Title: Superficial saphenofemoral arteriovenous fistula as hemodialysis access: clinical experience

Authors:

João A Corrêa (cdh.fsp@gmail.com)
Luiz Carlos Abreu (luizcarlos@usp.br)
Vitor E Valenti (valenti@unifesp.br)
Tatiana D de Carvalho (carvalho.td1@gmail.com)
Fausto Miranda Júnior (cdh.fsp@usp.br)

Version: 2 Date: 16 March 2010

Author’s response to reviews: see over
Dear Dr. Markus Buchler,

We enclose the research article entitled “Superficial saphenofemoral arteriovenous fistula as hemodialysis access: clinical experience”, by João Antônio Corrêa, Vitor E. Valenti, Luiz Carlos de Abreu, Tatiana Dias de Carvalho and Fausto Miranda Junior to submission to BMC Surgery.

The material has not been published elsewhere and is being submitted only to BMC Surgery. The authors and coauthors also accept full responsibility for the accuracy of all content, release and assign all rights for the publication of the manuscript to BMC Surgery and state that there are no conflicts of interest related to the manuscript. Authorship has been granted to the financial support from Núcleo de Estudos, Pesquisas e Assessoria à Saúde da Faculdade de Medicina do ABC (NEPAS-FMABC) and to those individuals who have contributed substantially to the manuscript.

With our best wishes,

João Antônio Corrêa
Vitor E. Valenti
Luiz Carlos de Abreu
Tatiana Dias de Carvalho
Fausto Miranda Junior